Author's response to reviews

Title: HIV/AIDS knowledge in detention in Hunan province, China

Authors:

weidong zhang (weidongch@yahoo.de)
xinya wang (sophia-xinya@163.com)
xi chen (chenxi161@sohu.com)
fan lv (fanlv@vip.sina.com)

Version: 5 Date: 24 April 2010

Author's response to reviews: see over
Dear editors,

On behalf of my coauthors I send you enclosed a revised manuscript entitled "HIV/AIDS knowledge in detention in Hunan province, China" that we want to submit for publication as a research article in BMC Public Health.

The article under number MS: 1269255328278881, we are very happy to hear from you that our article is ready for publication; we revised it finally and want to send it for publication.

This work has not been published in this or a substantially similar form [in print or electronically, including on a web site], nor accepted for publication elsewhere, nor is it under consideration by another publication. All authors have read and approved of the manuscript. None of the authors has any conflicts of interest.

We hope you find our manuscript suitable for publication in your journal.

With kind regards,
Weidong Zhang